Author's response to reviews

Title: A Review of the Methodological Features of Systematic Reviews in Maternal Medicine

Authors:

Lumaan Sheikh (lumaan.sheikh@aku.edu)
Shelley Johnston (SNH005@bham.ac.uk)
Shakila Thangaratinam (s.thangaratinam@bham.ac.uk)
Mark D Kilby (kilbymd@adf.bham.ac.uk)
Khalid S Khan (k.s.khan@bham.ac.uk)

Version: 4 Date: 26 February 2007

Author's response to reviews: see over
26th February 2007

To
Editor
BioMed Central

Dear Editor

Re: A Review of the Methodological Features of Systematic Reviews in Maternal Medicine

Many thanks for the acceptance of our manuscript for publication in BioMed Central. As per the editorial board’s recommendation we would like our manuscript to be published in BMC Medicine.

We have addressed the points raised by the editor in this final version that includes provision of checklist as Table 1 and detailed explanation of the use of $ sign in the methods section.

We look forward to hearing from you.

Yours Sincerely
Dr Shakila Thangaratinam MRCOG
Clinical Lecturer in Obstetrics and Gynaecology and Clinical Epidemiology
Academic Unit
3rd floor
Birmingham Women's Hospital
Birmingham B15 2TG, UK
E Mail: s.thangaratinam@bham.ac.uk
         shakila@doctors.org.uk
Ph: 07887775891       Fax: +44 (0)121 4154837